SGI Workstations (96 Total):

W20-575 (25) 4-035^y (20) 2-225 (13) 1-142 (10) 66-080 (10) Rotch Library ( 8) 37-312 ( 4) 11-113 ( 3) 11-116 ( 3)

y = reservable, but otherwise public

More SGI machines are currently being installed in 10-250 and 3-133. The public combination is currently 25048*, but it will change in October. The command cview should give a list of machines in public clusters and how many are available at the moment.

SGI Workstations (96 Total):

W20-575 (25) 4-035^y (20) 2-225 (13) 1-142 (10) 66-080 (10) Rotch Library ( 8) 37-312 ( 4) 11-113 ( 3) 11-116 ( 3)

y = reservable, but otherwise public

More SGI machines are currently being installed in 10-250 and 3-133. The public combination is currently 25048*, but it will change in October. The command cview should give a list of machines in public clusters and how many are available at the moment.

SGI Workstations (96 Total):

W20-575 (25) 4-035^y (20) 2-225 (13) 1-142 (10) 66-080 (10) Rotch Library ( 8) 37-312 ( 4) 11-113 ( 3) 11-116 ( 3)

y = reservable, but otherwise public

More SGI machines are currently being installed in 10-250 and 3-133. The public combination is currently 25048*, but it will change in October. The command cview should give a list of machines in public clusters and how many are available at the moment.

SGI Workstations (96 Total):

W20-575 (25) 4-035^y (20) 2-225 (13) 1-142 (10) 66-080 (10) Rotch Library ( 8) 37-312 ( 4) 11-113 ( 3) 11-116 ( 3)

y = reservable, but otherwise public

More SGI machines are currently being installed in 10-250 and 3-133. The public combination is currently 25048*, but it will change in October. The command cview should give a list of machines in public clusters and how many are available at the moment.

SGI Workstations (96 Total):

W20-575 (25) 4-035^y (20) 2-225 (13) 1-142 (10) 66-080 (10) Rotch Library ( 8) 37-312 ( 4) 11-113 ( 3) 11-116 ( 3)

y = reservable, but otherwise public

More SGI machines are currently being installed in 10-250 and 3-133. The public combination is currently 25048*, but it will change in October. The command cview should give a list of machines in public clusters and how many are available at the moment.

SGI Workstations (96 Total):

W20-575 (25) 4-035^y (20) 2-225 (13) 1-142 (10) 66-080 (10) Rotch Library ( 8) 37-312 ( 4) 11-113 ( 3) 11-116 ( 3)

y = reservable, but otherwise public

More SGI machines are currently being installed in 10-250 and 3-133. The public combination is currently 25048*, but it will change in October. The command cview should give a list of machines in public clusters and how many are available at the moment.

SGI Workstations (96 Total):

W20-575 (25) 4-035^y (20) 2-225 (13) 1-142 (10) 66-080 (10) Rotch Library ( 8) 37-312 ( 4) 11-113 ( 3) 11-116 ( 3)

y = reservable, but otherwise public

More SGI machines are currently being installed in 10-250 and 3-133. The public combination is currently 25048*, but it will change in October. The command cview should give a list of machines in public clusters and how many are available at the moment.

SGI Workstations (96 Total):

W20-575 (25) 4-035^y (20) 2-225 (13) 1-142 (10) 66-080 (10) Rotch Library ( 8) 37-312 ( 4) 11-113 ( 3) 11-116 ( 3)

y = reservable, but otherwise public

More SGI machines are currently being installed in 10-250 and 3-133. The public combination is currently 25048*, but it will change in October. The command cview should give a list of machines in public clusters and how many are available at the moment.

SGI Workstations (96 Total):

W20-575 (25) 4-035^y (20) 2-225 (13) 1-142 (10) 66-080 (10) Rotch Library ( 8) 37-312 ( 4) 11-113 ( 3) 11-116 ( 3)

y = reservable, but otherwise public

More SGI machines are currently being installed in 10-250 and 3-133. The public combination is currently 25048*, but it will change in October. The command cview should give a list of machines in public clusters and how many are available at the moment.